Author's response to reviews

Title: Chylous ascites associated with chylthorax; a rare sequela of penetrating abdominal trauma: a case report.

Authors:

Joseph M Plummer (joseph_plummer@yahoo.com)
Michael E McFarlane (michael.mcfarlane@uwimona.edu.jm)
Archibald H McDonald (archibald.mcdonald@uwimona.edu.jm)

Version: 6 Date: 6 October 2007

Author's response to reviews: see over
October 4, 2007

The Editor,
Journal of Medical Case Reports

Dear Sirs,

We hereby re-submit for publication the manuscript entitled ‘Chylos ascites associated with chylothorax; a rare sequela to penetrating abdominal trauma: a case report’.

We have addressed the comments by the reviewer regarding the pneumothorax and its treatment in the case report. This is addressed in the case report’s first paragraph.

In addition minor improvements in the use of English have been made throughout the document. We hope the manuscript now meets the Journal’s specifications for publication.

We patiently await your decision.

Yours truly,

Joseph Plummer